Dear Editors and Reviewers:

Thank you for your letter and the reviewers’ comments concerning our manuscript entitled “Treatment of extracranial internal carotid artery dissecting aneurysm with SUPERA stent implantation: A report of two case” (ID: 69901). These comments were helpful in revising and improving our paper, and were valuable for our study. We have examined your comments carefully and revised the manuscript based on your suggestions. We hope that we have addressed your concerns. We have provided a point by point response to the reviewer’s comments below.

Reviewer #Company Editor-in-Chief

Reviewer’s comment  (I have reviewed the Peer-Review Report, full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office's comments and the Criteria for Manuscript Revision by Authors. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.)

Response: In the revised version, we uploaded figures made in POWERPOINT.

Thank you and best regards.

QIU Minjian

2021-10-26